Author's response to reviews

Title: The prognostic role of CD133 expression in colorectal cancer: a meta-analysis

Authors:

Ke Wang (blwangke@126.com)
Jianjun Xu (jianjunxubl@126.com)
Jian Huang (drhuangjian@zju.edu.cn)
Junshu Zhang (blzhjunshu@126.com)

Version: 2 Date: 19 August 2012

Author's response to reviews: see over
Dear Nette,

RE: Submission ID: 9158779678617237

We would like to thank BioMed Central for giving us the opportunity to revise our manuscript, and also thank your suggestful comments. In this revised version, the concerns of you and your helpful suggestions for improvement have been addressed. Here we provide a point-by-point response to these comments as below. We are looking forward to hearing from you.

Sincerely,

Junshu Zhang
Department of Surgery
Linan People’s Hospital
Hangzhou 311300
Tel: +86-571-87315009
Fax: +86-571-87022776
E-mail: blzhjunshu@126.com
Q1: Please adhere to the PRISMA guidelines.

A1: According to your instructions, we have added PRISMA to the manuscript.

Q2: Please describe in details the meta-analysis.

A2: We thank your suggestion. In the revised text, we have described in details in materials and methods.

Q3: Requesting copyediting before review.

A3: According to your instructions, the quality of written English has been improved in the revised version.